Author's response to reviews

Title: Hyperadiponectinemia enhances osteogenesis through osteoblast formation in mice

Authors:

Yasuhiro Mitsui (mitsui_yasuhiro@kurume-u.ac.jp)
Masafumi Gotoh (gomasa@med.kurume-u.ac.jp)
Nobuhiro Fukushima (fukushima_nobuhiro@kurume-u.ac.jp)
Isao Shirachi (shirachi_isao@kurume-u.ac.jp)
Shuichi Otabe (otabeshu@yahoo.co.jp)
Xiaohong Yuan (yuan@med.kurume-u.ac.jp)
Toshihiko Hashinaga (hashinaga_toshihiko@kurume-u.ac.jp)
Nobuhiro Wada (wada_nobuhiro@kurume-u.ac.jp)
Akiko Mitsui (harukamama_m_921@yahoo.co.jp)
Tatsuhiko Yoshida (yoshida_tatsuhiro@kurume-u.ac.jp)
Shiro Yoshida (marine7blue@hotmail.com)
Kentaro Yamada (yamada@med.kurume-u.ac.jp)
Kensei Nagata (knspine@med.kurume-u.ac.jp)

Version: 2 Date: 15 June 2010

Author's response to reviews:

Editorial Assistant BMC Series Journals
To whom it may concern.

Thank you for your comments.
We have revised the manuscript in accordance with your suggestions.

Yours sincerely,

Yasuhiro Mitsui, M.D.